Cover letter to Dr. Stamm

Dear Dr. Stamm,

Again we send to you our revised version according to the recommendations of Dr. Kirsi Sipilä.

I want to point on the fact that again an American native speaker re-read the manuscript so that the text is now the result of the corrections of 2 (!) native speakers.

We included the points of Dr. Sipilä a second time and we marked it in the text in red again. Small parts of the text are deleted according to the recommendation of Dr. Sipilä (statistics).

I hope that our work can now be accepted this form for publication in H&F Medicine.

With regards

Prof. Dr. Rainer Laskawi
University of Göttingen